Author's response to reviews

Title: Effects of light deprivation on visual evoked potentials in migraine without aura

Authors:

Gianluca Coppola (gianluca.coppola@gmail.com)
Julien Crémers (julien.cremers@ulg.ac.be)
Pascale Gérard (pasgerard@hotmail.com)
Francesco Pierelli (fpierelli@gmail.com)
Jean Schoenen (jschoenen@ulg.ac.be)

Version: 3 Date: 24 July 2011

Author's response to reviews:

MS: 6003533955435611
Effects of light deprivation on visual evoked potentials in migraine without aura

BMC Neurology

Major revisions (we require the author to make these changes)
Structure: Please include conclusion section after discussion.
Done.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)
Highlighting/tracking: Please remove from the manuscript.
Done

Structure: Please place figure legends after reference list.
Done

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Checked

Best regards
Gianluca Coppola